Author's response to reviews

Title: Trials and Tribulations of Recruiting 2,000 Older Women onto a Clinical Trial Investigating Falls and Fractures: Recruitment and Methodology of the Vital D Study

Authors:

Kerrie M Sanders (kerrie@barwonhealth.org.au)
Amanda L Stuart (amandh@barwonhealth.org.au)
Elizabeth N Merriman (elizabe@barwonhealth.org.au)
Meaghan L Read (meaghan.read@bms.com)
Mark A Kotowicz (markk@barwonhealth.org.au)
Doris Young (d.young@unimelb.edu.au)
Roderick Taylor (gmc@gpageelong.com.au)
Ian Blair-Holt (ian@cvmp.com.au)
Alistair G Mander (agmander@gmail.com)
Geoffrey C Nicholson (geoffn@barwonhealth.org.au)

Version: 3 Date: 27 October 2009

Author's response to reviews: see over
27th October 2009

Dear Editor,

Re: MS: 5299756672827026

Trials and Tribulations of Recruiting 2,000 Older Women onto a Clinical Trial Investigating Falls and Fractures: Recruitment and Methodology of the Vital D Study

We note that the manuscript was well regarded by both reviewers and have addressed the minor issues mentioned by one of the reviewers.

We hope the manuscript is now ready for publication,

A/Prof Kerrie Sanders (Corresponding author)
Head, Clinical Research Unit
Department of Clinical and Biomedical Sciences; Barwon Health
The University of Melbourne
PO Box 281
Geelong, Victoria 3220

Phone +61 3 5226 7834 and (m) +61411 149 789
Email: kerrie@barwonhealth.org.au

Reviewer 2
Areas for clarification

1. The ‘design and intervention’ from the abstract and background.
2. The typing error has been corrected
3. The sentence in the discussion has been shortened and clarified as follows: "The ‘bring a friend’ morning tea offered to participants in the early stages of recruitment was also a worthwhile activity. The morning tea resulted in 30 new participants from the 200 invited participants.”